Author’s response to reviews

Title: Spectrum of CFTR mutations in Chechen cystic fibrosis patients: high frequency of c.1545_1546delTA (p.Tyr515X; 1677delTA) and c.274G>A (p.Glu92Lys, E92K) mutations in North Caucasus.

Authors:
Nika Petrova (npetrova63@mail.ru)
Nataliya Kashirskaya (kashirskayanj@mail.ru)
Dzhamilya Saydaeva (saydaeva78@mail.ru)
Alexander Polyakov (polyakov@med-gen.ru)
Tagui Adyan (tagui.adyan@yandex.ru)
Olga Simonova (oisimonova@mail.ru)
Yulia Gorinova (ygorinova@yandex.ru)
Elena Kondratyeva (elenafpk@mail.ru)
Victoria Sherman (tovika@yandex.ru)
Olga Novoselova (lgnovoselova@gmail.com)
Tatiyana Vasilyeva (vasilyeva_debrie@mail.ru)
Andrey Marakhonov (marakhonov@gmail.com)
Milan Macek Jr. (milan.macek.jr@lfmotol.cuni.cz)
Evgeny Ginter (ekginter@mail.ru)
Rena Zinchenko (renazinchenko@mail.ru)

Version: 3 Date: 12 Mar 2019

Author’s response to reviews:

Dear Editor,

Thank you for your comments.
I tried to make all the changes that you've asked to do.

1. Please state in the cover letter whether the map depicted in figure 1 is your own or taken from another source.

The answer: the information is added on page 8.

Figure: Ethnolinguistic groups in the Caucasus region. Wikimedia Commons is a collection of 49,714,663 freely usable media files to which anyone can contribute. https://commons.wikimedia.org/w/index.php?curid=2430263. Accessed 09 Jan 2019. This media licensed under the Creative Commons Attribution 2.5 License: https://creativecommons.org/licenses/by/2.5/.

2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

The answer: all is done on page 6.

3. Please rename Figure, tables, and additional files to Figure Legends.

The answer: all is done on page 8.

Best regards,

Nataliya Kashirskaya